Author's response to reviews

Title: Use of complementary and alternative medicines for infertility enhancement by a sample of Turkish women: a descriptive study

Authors:

Tamer Edirne (tameredirne@yahoo.com)
Secil G Arica (secilgunher@hotmail.com)
Sebahat Gucuk (sebahatgu@yahoo.com)
Recep Yildizhan (recepyildizhan@yahoo.com)
Ali Kolusari (dralikolusari@yahoo.com)
Ertan Adali (ertanadali@yahoo.com)
Muhammet Can (mcan29@yahoo.com)

Version: 4 Date: 9 March 2010

Author's response to reviews:

Reviewer's report
Title: Use of complementary and alternative medicines for infertility enhancement by a sample of Turkish women: a descriptive study
Version: 3 Date: 8 March 2010
Reviewer: Osman Günay
Reviewer's report:
Dear Authors,

There are some mistakes in Table 1. I couldn't understand that 5 year for education and 377 USD for monthly income were in which category.

Responses: The mistakes are revised and the mentioned categories are written in more detail.

Level of interest: An article whose findings are important to those with closely related research interests
Quality of written English: Acceptable
Statistical review: No, the manuscript does not need to be seen by a statistician.